Author's response to reviews

Title: IL-13 expression by blood T cells and not eosinophils is increased in asthma compared with non-asthmatic eosinophilic bronchitis

Authors:

Salman Siddiqui (saman95@yahoo.com)
Glenn Cruse (gpc11@le.ac.uk)
Susan McKenna (susanmckenna@btopenworld.com)
William Monteiro (wrm2@le.ac.uk)
Vijay Mistry (vm53@le.ac.uk)
Andrew Wardlaw (aw24@le.ac.uk)
Christopher Brightling (ceb17@le.ac.uk)

Version: 4 Date: 18 May 2009

Author's response to reviews:

Editor-in-Chief
BMC Pulmonary Medicine

18th May 2009

Dear Editor,

Re: MS-3068455226306878 IL-13 expression by blood T cells and not eosinophils in asthma compared to non-asthmatic eosinophilic bronchitis

Thank you for giving us the opportunity to resubmit a revised manuscript. We have reformatted the figures as requested by the referee and amended the manuscript where necessary to conform to your journal style. We hope that you consider it now suitable for publication in your journal.

Yours sincerely,

Chris Brightling